Author's response to reviews

Title: Subhepatically located appendix due to adhesions and foetal malrotation: two case reports

Authors:

Joseph Y S Ting (Joseph.Ting@mater.org.au)
Rebecca Farley (Rebecca.Farley@mater.org.au)

Version: 6 Date: 20 June 2008

Author's response to reviews:

In response to feedback from Dr D Lobo regarding uncertain cause of right abdominal pain and its impact on clinical outcome for Case 2, I ahve added a clarifying sentence at the end of the final paragraph on p 5, "The patient presumably had acute colitis with pericolonic inflammation that settled with IV antibiotics." This is highlighted in pink.